Author’s response to reviews

Title: uPA is upregulated by high dose celecoxib in women at increased risk of developing breast cancer

Authors:

  - Wenyi Qin (wqin@medicine.nodak.edu)
  - Weizhu Zhu (wzhu@medicine.nodak.edu)
  - John E Hewett (hewettj@health.missouri.edu)
  - George Rottinghaus (rottinghausg@missouri.edu)
  - Yin-chieh Chen (yinchieh_chen@yahoo.com)
  - John T Flynn (john.flynn@jefferson.edu)
  - Beth Kliethermes (bkliethermes@medicine.nodak.edu)
  - Ferdinando Mannello (ferdinando.mannello@uniurb.it)
  - Edward R Sauter (esauter@medicine.nodak.edu)

Version: 4 Date: 29 September 2008

Author’s response to reviews:

September 29, 2008

Gentlemen:

Attached please find the latest version of the manuscript which includes the changes requested.

Sincerely,

Edward Sauter